Author’s response to reviews

Title: Genes CEP55, FOXD3, FOXF2, GNAO1, GRIA4, and KCNA5 as potential diagnostic biomarkers in colorectal cancer

Authors:

Nina Hauptman (nina.hauptman@mf.uni-lj.si)
Daša Skok (dasajevsinekskok@kis.si)
Elena Spasovska (elena.spasovska90@gmail.com)
Emanuela Boštjančič (emanuela.bostjancic@mf.uni-lj.si)
Damjan Glavač (damjan.glavac@mf.uni-lj.si)

Version: 2 Date: 29 Mar 2019

Author’s response to reviews:

Dear Editor,

I am responding to you comments here.

1. We have renamed the ‘Materials and Methods’ to ‘Methods’ and moved it after the Background.

2. We have checked the manuscript for formatting required by your journal.

3. We have included that statement that all the author have read and approved the manuscript. I have personally sent email again to all authors regarding this topic.

4. We have revised the Declaration section and reordered it.
5. We are aware of the copyright permission to reproduce figures or tables previously published elsewhere. We do not require permission for any of the figures. All the figures were produced by the authors with software, which is now indicated in the figure legends. We have to mention that the software used to produce the figures is described in the Methods sections, subsection 'Data visualization, text mining and survival analysis'. Figure 2 - heatmap was produced with ComplexHeatmap library in R programming language, and Figure 3 was produced by STRING web-application. Figures 1 and 4 were produced in Microsoft Office software. I have added the software used in figure captions as you desired, but I have not seen software stated in any of the other articles in your journal.

6. We have included the role of the funding body in Declaration section.

7. We have uploaded files as described.